Author's response to reviews

Title: The impact of single versus mixed schistosome species infections on liver, spleen and bladder morbidity within Malian children pre- and post-praziquantel treatment

Authors:

Artemis Koukounari (artemis.koukounari@imperial.ac.uk)
Christl A. Donnelly (c.donnelly@imperial.ac.uk)
Moussa Sacko (msacko@afribonemali.net)
Adama D. Keita (gadkeita@hotmail.com)
Aly Landouré (aly_landoure@hotmail.com)
Robert Dembelé (rdembele2000@yahoo.fr)
Elisa Bosqué-Oliva (e.bosque-oliva@imperial.ac.uk)
Albis F Gabrielli (GabrielliA@who.int)
Anouk Gouvras (a.gouvras06@imperial.ac.uk)
Mamadou Traoré (traorem@afribonemali.net)
Alan Fenwick (a.fenwick@imperial.ac.uk)
Joanne P. Webster (joanne.webster@imperial.ac.uk)

Version: 7 Date: 28 July 2010

Author's response to reviews:

Dear Sir or Madam,

Thanks very much for your message. We have now addressed the pending changes that needed to be made prior to publication. The numbering of the references has changed accordingly in the revised manuscript.

With Kind Regards,

Artemis Koukounari on behalf of all co-authors